CS 66.460 Introduction to Graphical Human{Machine Interfaces Spring 1997 VIDEO LAB REPORT

Your name: Today’s date:

1. List the systems and/or topics which were presented on the tape(s) shown today. For which user community is each intended?

2. What were the innovative or most important features in the author(s) opinion?

3. What were the innovative or most important features in your opinion?

4. List any disadvantages or unclear issues or unintelligible features. Did the author(s) acknowledge these deficiencies?

5. If you were designing and implementing similar system(s), what might you do differently?

Why? How?

6. THE BOTTOM LINE: What is your overall impression of the work shown on the tape(s) you just saw?